Author's response to reviews

Title: Effects of preoperative feeding with a whey protein plus carbohydrate drink on the acute phase response and insulin resistance. A randomized trial.

Authors:

Francine Perrone (francineper@uol.com.br)
Antônio C da Silva-Filho (antoniocordeirosf@hotmail.com)
Isa F Adôrno (isafelixadorno@yahoo.com.br)
Nadia T Anabuki (anabukitomiko@hotmail.com)
Fernando S Leal (frnsleal@hotmail.com)
Tariane Colombo (taricol@hotmail.com)
Benedito D da Silva (doliveira@ufmt.br)
Diana B Dock-Nascimento (dianadock@hotmail.com)
Aderson Damião (aderson_damiao@uol.com.br)
José E de Aguilar-Nascimento (aguilar@terra.com.br)

Version: 6 Date: 12 June 2011

Author's response to reviews:

Dear Editor

A few changes were done to this final presentation of our manuscript. Another author was added in the manuscript and also in the system accordingly.

I hope now we are all set

Best regards

JE Aguilar-Nascimento, MD, PhD